Author's response to reviews

Title: Qingfei Xiaoyan Wan alleviates asthma through multi-target network regulation

Authors:

Zhenying Zhao (z.zhenying@gmail.com)
Yingbo Miao (miaoyingbo@126.com)
Pengwei Pan (nkuill@gmail.com)
Binfeng Cheng (chengbinfeng@mail.nankai.edu.cn)
Zhiqiang Lu (luzq85@163.com)
Gang Bai (gangbai@nankai.edu.cn)
Hong Wu (whmedu@126.com)

Version: 7 Date: 10 July 2013

Author's response to reviews: see over
Dear Dr. Rowles,

Thanks for your comments on the ethics statement and reviewers’ satisfaction about our MS (ID: 2145680990952720). Herein, we want to make it clear that all experiments were carried out according to the Guide for the Care and Use of Experimental Animals. Studies were approved by the Institute Committee of the Animal Care of Nankai University, China.

In the last version (June 26), we applied to add Zhiqiang Lu as another author. Because of some situation, now we quit that application and just keep the primary authorship. Sorry for any inconvenience.

**Changes made in the manuscript:**
(Revised portion are marked in red in the paper.)

<table>
<thead>
<tr>
<th>Line 119, page 6</th>
<th>Revisions</th>
</tr>
</thead>
<tbody>
<tr>
<td>The animals were housed at 22 ± 2 °C with 55 ± 10 % humidity, 12 h light/dark cycle, and had free access to species-specific food and tap water. All experiments were carried out according to the Guide for the Care and Use of Experimental Animals. Studies were approved by the Institute Committee of the Animal Care of Nankai University, China.</td>
<td></td>
</tr>
</tbody>
</table>

Thanks again for your sincere work. Looking forward to hearing from you.

With regards.

Yours Sincerely,
Hong Wu with coauthors